Author's response to reviews

Title: Testing assumptions for endophenotype studies in ADHD: Reliability and validity of tasks in a general population sample

Authors:

Jonna Kuntsi (j.kuntsi@iop.kcl.ac.uk)
Penny Andreou (p.andreou@iop.kcl.ac.uk)
Jonathan Ma (spigjma@iop.kcl.ac.uk)
Norbert A Borger (n.a.borger@rug.nl)
Jaap J van der Meere (jj.van.der.meere@rug.nl)

Version: 3 Date: 26 September 2005

Author's response to reviews: see over
26 September 2005

Dear BioMed Central Editorial Team

Re: ‘Testing assumptions for endophenotype studies in ADHD: Reliability and validity of tasks in a general population sample’

Thank you for confirming the acceptance of our manuscript for publication in BMC Psychiatry.

We have made the requested formatting changes as follows:

**Title & Abstract**
We confirm that these are identical in the manuscript and in the submission system.

**Author list**
The list of authors is written exactly as it appears in the submission system.

**Section headings**
All unnecessary capitals have been removed.

**Introduction**
This section has been renamed ‘Background’.

**Declaration of competing interests**
This section has been renamed ‘Competing interests’

**Tables**
Roman Numerals have been replaced by standard numbers (including the Results section).

We are delighted that our manuscript has been accepted for publication in BMC Psychiatry and look forward to seeing it in print soon.

Yours sincerely

Jonna Kuntsi PhD
Lecturer, MRC SGDP Centre, Institute of Psychiatry